
NSF CC-NIE Working Group Meetng 

June 16, 2014 

Agenda 

 

Time: 3:30-4:30pm, June 16, Monday 

Locaton: Physics building, Conference room 403B  

Agenda: 

1. Status of measurement infrastructure  

2. Update from CEN  

3. Status of Science-DMZ equipment at Storrs 

4. Update from UCHC  

5. Next step 

6. Open discussion on other issues 

Minutes  
Present: 

UConn: Bob, Jef 

UCHC: Jef D., Peter (O’Neill) 

1.   Status of measurement infrastructure    
A. Jef F and Reynaldo will speak Friday and go over details of Reynaldo’s modifcatons 
B. UC Storrs and UCHC will start using perfsonar between locatons.  Need to coordinate. 
 
2. Update from CEN  
No update. 
 
3. Status of Science-DMZ equipment at Storrs 
       A. Bob is working with the vendor to generate the correct quotes with the correct parts, this is going 
to take a few iteratons to get it correct. The most recent quote was 70-90k higher than the original due 
to newer modules as well as an additonal 10G line card.  We expect the cost to go down once all is 
corrected. 
 
4. Update from UCHC  



       A. Perfsonar is up and running 
       B. OSG issues around CONDOR Jef (UCHC) is working on these issues 
 
5. Next step 
        1. Perfsonar between UCHC and UC Storrs 
        2. Finalize Quotes 
        3. Discuss confguraton and installaton of local switch gear for R. Jones 
                Jef F needs to check on fber installaton quotes 
        4. Request that Bing provides a budget status update for next meetng. 
 
6. Open discussion on other issues 
        Dell networking - The group agreed that we are not interested in pursuing this at this tme 
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